Author's response to reviews

Title: Reliability of clinical tests to evaluate nerve function and mechanosensitivity of the upper limb peripheral nervous system

Authors:

Annina B Schmid (a.schmid@uq.edu.au)
Florian Brunner (florian.brunner@balgrist.ch)
Luomajoki Hannu (hannu@physios.ch)
Held Ulrike (ulrike.held@usz.ch)
Künzer Sabine (sabine.kuenzer@balgrist.ch)
Bachmann M Lucas (lucas.bachmann@usz.ch)
Coppieters W Michel (m.coppieters@uq.edu.au)

Version: 4 Date: 8 January 2009

Author's response to reviews: see over
Article submission

Dear Dr. Norton

We would like to thank the Editor and the Editorial Board for accepting our article for publication in BMC Musculoskeletal Disorders.

We have addressed the change in the sentence as suggested by Reviewer 3. The correct sentence now reads as follows (page 15, paragraph 2):

**However, the literature on the validity of bedside neurological examination in patients with musculoskeletal symptoms is scarce and there are no studies available which examine the validity of nerve palpation.**

Additionally, we have formatted the manuscript according to the journal style.

Thank you very much for publishing our article.

Yours truly,

Annina Schmid

On behalf of the co-authors:

Florian Brunner Hannu Luomajoki,
Sabine Küenzer,
Michel Coppieters

Ulrike Held
Lucas M. Bachmann